Author's response to reviews

Title: Extramedullary haematopoiesis presented as intrathoracic tumour in a patient with alpha-thalassaemia

Authors:

Dmitry Bobylev (dmitry_bobylev@yahoo.de)
Ruoyu Zhang (Zhang.ruoyu@mh-hannover.de)
Axel Haverich (Haverich.axel@mh-hannover.de)
Marcus Krueger (Krueger.marcus@mh-hannover.de)

Version: 2 Date: 21 April 2013

Author's response to reviews: see over
Dear Dr. Vipin Zamvar,

Thank you for consideration of our manuscript “Extramedullary haematopoiesis presented as intrathoracic tumour in a patient with alpha-thalassaemia” by Dmitry Bobylev and colleges for publication in your journal. We have changed the manuscript according to reviewer and your comments.

We do hope that this manuscript will be published in the Journal of cardiothoracic surgery.

Sincerely,

Dmitry Bobylev

Dmitry Bobylev, MD.
Department of Cardiothoracic, Transplantation, and Vascular Surgery,
Hannover Medical School
Carl-Neuberg Str. 1, 30625 Hannover, Germany
Tel: +49-511-532-9397
Fax: +49-511-532-9832
E-mail: dmitry.bobylev@yahoo.de